Author’s response to reviews

Title: Thermal damage width and hemostatic effect of bipolar electrocoagulation, LigaSure, and Ultracision techniques on goat mesenteric vessels and optimal power for bipolar electrocoagulation

Authors:

Jun Liang (13582026972@163.com)
Huimin Xing (18731160881@163.com)
Yali Chang (shanya596huangch@163.com)

Version: 1 Date: 23 Sep 2019

Author’s response to reviews:

Dear editor,

Thank you for your letter with regard to our manuscript entitled ‘Thermal damage width and hemostatic effect of bipolar electrocoagulation, LigaSure, and Ultracision techniques on goat mesenteric vessels and optimal power for bipolar electrocoagulation’ again. We greatly appreciate for your time in reviewing our manuscript. Accordingly, we highlight all the updates in the manuscript by using the track changes mode. The point-by-point answers to the comments are listed as below, and we hope it can be acceptable for publication in the journal.

Assistant Editor Comments:

1. Euthanasia

Please include the pentobarbital dosage under ‘methods’.

Response: Firstly, thank you very much for your recognition and encouragement of our research, and your suggestions will make the research more valuable and perfect. We are sorry for the lacking content, and we have added it.

2. Availability of Data and Materials
The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

Response: Thank you for your advice. And we have revised it according to instructions above.

3. Section

When re submitting please upload your manuscript to the ‘general surgery‘ section.

Response: OK.

4. ARRIVE

Please remove the ARRIVE checklist from your file inventory as this has been checked and is not required at this stage in the editorial process.

Response: OK.

5. Formatting

- Please remove the language editing certificate and the authorship change form as they are not required at this stage in the editorial process.

Response: OK.

-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Response: OK.
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: OK.